Author's response to reviews

Title: Effectiveness and cost-effectiveness of an exposure-based return-to-work programme for patients on sick leave due to common mental disorders: design of a cluster-randomized controlled trial

Authors:

Erik Noordik (f.w.noordik@amc.nl)
Frank J. Van Dijk (f.j.vandijk@amc.nl)
Karen Nieuwenhuijsen (k.nieuwenhuijsen@amc.nl)
Jac J.L. Van der Klink (j.j.l.van.der.klink@med.umcg.nl)

Version: 3 Date: 10 May 2009

Author's response to reviews: see over
Dear editor,

All major and minor revisions you requested for are made in the manuscript (MS 8947740522530819):

**Major revision**
Reference 34: We provided full details;

**Minor revisions**
Title page: We removed the words ‘Study design’ on top of the page;

Abstract: the trial registration number is mentioned in the submission system as well as the manuscript;

Methods: ‘Methods’ is change into ‘Methods/design’ in both the Abstract and the manuscript;

Table and figure placement text: text is removed;

Figure legends: the heading ‘Figure legends’ is added to the top of this section;

Figure cropping: the figure is cropped as closely as possible.

Yours sincerely,

Erik Noordik